Author's response to reviews

Title: Increased CCL19 and CCL21 levels promote fibroblast ossification in ankylosing spondylitis hip ligament tissue

Authors:

Yang Qin (yangqinsus@126.com)
Li Da He (bzmcldh@163.com)
Zhou Jian Sheng (zhoujiasheng0902@163.com)
Miao Ming Yong (miaomy@163.com)
Yang Sheng Sheng (Ysscn@126.com)
Xu Wei Dong (xuwdsanghai@126.com)

Version: 5 Date: 11 August 2014

Author's response to reviews:

Dear editors,

I have revise some aspects of my article such as simplifying the Introduction and revising image format. And most part is not modified.

I have answered the reviews questions one by one in the cover letter.

The Ethics Support contains:

1. Name of approved: Shanghai Changhai Hospital Ethics Committee
2. Ethics Committee approval document numbers: CHEC2013-194
3. International General registration number: U1111-1151-6685